ECONOMIA PUBBLICA The Italian Journal of Public Economics

Le imposte sulla prima casa in Italia, un equilibrio difficile fra decentramento e redistribuzione, di Giovanna Messina, Marco Savegnago

L'impatto della contabilità euro-compatibile in un'auspicabile evoluzione del Patto di Stabilità Interno, di Leonzio Rizzo, Massimiliano Ferraresi

Quanto conta l'effetto scuola nel ciclo primario? L'efficacia delle istituzioni scolastiche in Toscana, di Enrico Conti, Silvia Duranti, Carla Rampichini, Nicola Sciclone

Spesa energetica e competitività delle imprese italiane, di Ivan Faiella e Alessandro Mistretta

The Evaluation of Standard Expenditure Needs of Municipalities: the Case of Social Care Services in Italy, di Francesco Porcelli

Osservatorio Utilitatis

Il processo di riforma della disciplina in materia di contratti pubblici, di Pina Russo

Nuova serie a cura di







anno XLII - n. 3 - 2015

Direttore/Editor: Lorenzo Robotti

Direttore Responsabile/Managing Editor: Stefano Angeli

Comitato scientifico/Scientific committee: Lorenzo Bardelli, Alberto Biancardi, Giuseppe Bognetti, Claudio De Vincenti, Fabio Fiorillo, Massimo Florio, Silvia Giannini, Ernesto Longobardi, Alessandro Petretto, Andrea Zatti, Alberto Zevi.

Comitato editoriale/Editorial board: Luca Cerioni, Luca Del Bene, Barbara Ermini, Valeria Garotta, Chiara Gigliarano, Michele Giuranno, Nicola Matteucci, Agnese Sacchi, Raffaella Santolini.

Redazione/Editorial office: Ciriec, Centro italiano di ricerche e d'informazione sull'economia pubblica, sociale e cooperativa. Sesto San Giovanni (Milano), villa Mylius, Largo Lamarmora - telefono e fax: 02 26.26.11.57 – email: ciriecmi@tin.it, l.robotti@univpm.it.

Amministrazione, distribuzione, abbonamenti: viale Monza 106, 20127 Milano, tel. 02.28.37.141; fax abbonamenti 02.26.141.958; e-mail: riviste@francoangeli.it

Abbonamenti – Per conoscere il canone d'abbonamento corrente, consultare il nostro sito www.francoangeli.it, cliccando sul bottone "Riviste", oppure telefonare al nostro Ufficio Riviste (02-2837141) o, ancora, inviare una e-mail a riviste@francoangeli.it indicando chiaramente il nome della rivista. Il pagamento potrà essere effettuato tramite assegno bancario, bonifico bancario, versamento su conto corrente, o con carta di credito.

L'abbonamento verrà attivato non appena giunta la notifica dell'avvenuto pagamento del canone.

L'opera, comprese tutte le sue parti, è tutelata dalla legge sui diritti d'autore. Sono vietate e sanzionate (se non espressamente autorizzate) la riproduzione in ogni modo e forma (comprese le fotocopie, la scansione, la memorizzazione elettronica) e la comunicazione (ivi inclusi a titolo esemplificativo ma non esaustivo: la distribuzione, l'adattamento, la traduzione e la rielaborazione, anche a mezzo di canali digitali interattivi e con qualsiasi modalità attualmente nota od in futuro sviluppata).

Le fotocopie per uso personale del lettore possono essere effettuate nei limiti del 15% di ciascun fascicolo dietro pagamento alla SIAE del compenso previsto dall'art. 68, commi 4 e 5, della legge 22 aprile 1941 n. 633. Le fotocopie effettuate per finalità di carattere professionale, economico o commerciale o comunque per uso diverso da quello personale, possono essere effettuate a seguito di specifica autorizzazione rilasciata da CLEARedi, Centro Licenze e Autorizzazioni per le Riproduzioni Editoriali (www.clearedi.org; e-mail autorizzazioni@clearedi.org).

In caso di copia digitale, l'Utente nel momento in cui effettua il download dell'opera accetta tutte le condizioni della licenza d'uso dell'opera previste e comunicate sul sito www.francoangeli.it.

Reg. Tribunale di Milano 14-5-1971, n. 194 – Poste Italiane Spa – Sped. in Abb. Post. – D.L. 353/2003 (conv. in L. 27/02/2004 n. 46) art. 1, comma 1, DCB Milano. Copyright © 2016 by FrancoAngeli srl, Milano – Stampa: Global Print srl, via degli Abeti 17/1, Gorgonzola, Milano.

III quadrimestre 2015 – Finito di stampare nel mese di aprile 2016

Numero 3, anno 2015

Giovanna Messina e Marco Savegnago Le imposte sulla prima casa in Italia, un equilibrio difficile fra decentramento e redistribuzione	pag.	5
Massimiliano Ferraresi e Leonzio Rizzo L'impatto della contabilità euro-compatibile in un'auspicabile evoluzione del Patto di Stabilità Interno	»	31
Enrico Conti, Silvia Duranti, Carla Rampichini e Nicola		
Quanto conta l'effetto scuola nel ciclo primario? L'efficacia delle istituzioni scolastiche in Toscana	»	59
Ivan Faiella e Alessandro Mistretta Spesa energetica e competitività delle imprese italiane	»	85
Francesco Porcelli The Evaluation of Standard Expenditure Needs of Municipalities: The Case of Social Care Services in Italy	»	123
OSSERVATORIO UTILITATIS		
Pina Russo Il processo di riforma della disciplina in materia di contratti pubblici	»	159

The Evaluation of Standard Expenditure Needs of Municipalities: The Case of Social Care Services in Italy

di Francesco Porcelli*

ABSTRACT

This paper is a policy report providing an overview of the recent Italian experience in the evaluation of the standard expenditure needs of municipalities (6702 elected local authorities of regions with ordinary statutes), discussing in detail the main technical choices and the final methodologies adopted in relation to the mainstream methods reported in the economic literature and developed by other countries. For simplicity, the paper is focused on the social care sector (which absorbs more than 20% of municipal current expenditure), the methodological approach being the same in the case of other local services.

JEL: H72, H77, H83

Keywords: standard expenditure needs, municipalities, Italy, social care

* Centre for Competitive Advantage in the Global Economy (University of Warwick, UK) and SOSE SpA. E-mail: fporcelli78@gmail.com.

SOSE Soluzioni per il Sistema Economico S.p.a. is a company, owned both by the Italian Ministry of Economy and Finance (88%) and Bank of Italy (12%), specialized in the provision of statistical and econometric analysis in any field of the private and public sector. Since 2011 SOSE, among its activities, elaborates the econometric models for the the evaluation of the standard expenditure needs of Italian local governments. This document does not necessarily reflect the official opinion of SOSE – Soluzioni per il Sistema Economico S.p.A. and commits only the author.

Economia Pubblica, numero 3, anno 2015

Copyright © FrancoAngeli

1. Introduction

Fiscal equalization, although absorbs roughly 5% of total public expenditure on average within OECD countries, represents a crucial aspect of all fiscal and administrative decentralization processes, since equalization grants limit the territorial imbalances that the devolution of a greater local autonomy tends to amplify. This aspect is of great importance in a country such as Italy, which is characterized by a marked economic duality between the north and the south of the country. The assessment of standard expenditure needs (SEN) for local authorities is, in turn, the main component of all fiscal equalization systems (Blöchliger *et al.*, 2007).

In the end of 2013, the Italian government produced the first wave of the assessment of SEN for 6702 municipalities. This marked the beginning of a radical reform of intergovernmental relations in Italy, taking the first and necessary step towards the construction of a new and more efficient mechanism for the distribution of equalization grants to finance the essential services of municipalities (34 billion euros in 2010).

The essential functions of municipalities regards twelve services: tax office (0.50 billion euros), technical office (1.02 billion euros), civil registry (0.55 billion euros), general services (6.39 billion euros), public roads (2.21 billion euros), local public transport (1.00 billion euros), land management and planning (1.67 billion euros), waste management (7.61 billion euros), general social services (4.67 billion euros), nursery services (1.44 billion euros), local police (2.64 billion euros), complementary education services (3.57 billion euros). For simplicity, the paper is focused on the social care sector, the methodological approach being the same in the case of other services.

The Italian law (Legislative Decree 216/10) assigns the task of processing methodologies for the determination of SEN for local authorities to SOSE Soluzioni per il Sistema Economico S.p.A. an independent company owned both by the Italian Ministry of Economy and Finance (88%) and Bank of

N.B: Copia ad uso personale. È vietata la riproduzione (totale o parziale) dell'opera con qualsiasi mezzo effettuata e la sua messa a disposizione di terzi, sia in forma gratuita sia a pagamento.

Italy (12%) specialized in the provision of statistical and econometric analysis in any field of the private and public sector. This choice follows a governance model that is considered by the OECD among the best suited for the management of equalization grants: a model that assigns technical/methodological tasks to an impartial body in order to facilitate mediation between the central government and local authorities¹.

The calculation of SEN is based on the idea that the financial needs of a local authority are an expression of the services provided, of the territorial features and of the social-economic and demographic characteristics of the resident population. By following this general approach, SOSE has developed an econometric methodology specific to the Italian system based on the Regression-based Cost Approach (RCA) methodology recognized by the international scientific community as the most advanced one (see SOSE, 2012, 2013). This methodology, which has been adopted in many countries including the United Kingdom and Australia, implies the determination of SEN through the estimation of a cost function and/or an expenditure function using the technique of multiple linear regression.

One of the main difficulties of calculating SEN using statistical techniques is the lack of information. In order to overcome this obstacle, the information provided by official sources (Budget Sheets, Ministry of the Interior, ISTAT, Ancitel, Ministry of Education, Land Registry Office, etc.) has been integrated with new data by sending all authorities a specific questionnaire for each standardized function. In this way a new database was built that, for the first time in Italy, allows a detailed analysis of outputs, inputs, methods of management and organizational decisions made in the production process of local services by local governments. The amount of data collected and processed for the determination of SEN, in addition to representing valuable information in itself, gives the valuation procedure a considerable degree of robustness. In addition, the survey of structural data through the questionnaire represents an extraordinary innovation in international techniques to evaluate SEN.

In the Italian case the law has prescribed a top-down approach for the valuation of SEN². This implies a two stage procedure. Firstly, the overall budget constraint (macro-budget) must be defined, that is the total amount of

² Concerning the financial implications of introducing SEN, note that article 1, paragraph 2 of Legislative Decree 216/10 states that the total amount of the SEN relating to fundamental functions for municipalities cannot exceed the present expenditure for the corresponding sectors.

Copyright © FrancoAngeli

¹ SOSE operates in collaboration with the Italian association of municipalities (ANCI) under the supervision of an independent governmental commission (COPAFF).

the equalization fund established by the central government. At a second stage, the resources must be allocated among local governments. Thus the valuation of SEN of local governments ends up in establishing a criterion for the distribution of a fixed amount of resources. SEN are primarily a relative concept. Even if the econometric procedure evaluates SEN in absolute monetary terms, this is functional to measuring the relative weight of any single service for any single local entity. Therefore, the entire procedure of calculation of the SEN ends up with the identification of a share in total expenditure, which can be named allotment coefficient³. In particular the total standard expenditure need of each municipality is composed of 12 allotment coefficients of expenditure (one for each function/service) and a comprehensive indicator, which is calculated as the ratio of the sum of the amount (in euro) of the SEN of all the essential services of the *i-th* municipality on the total standard expenditure of all municipalities.

In conclusion it is important to note that, although the gap between the standard and actual expenditure for each service provides a good reference point to judge the level of expenditure of each municipality, it is not a good indicator of local governments' efficiency in the provision of local services. The crude comparisons between standard expenditures and historic expenditures do not provide enough information to infer the ability of local governments and the effort they exert in the production of local services for two main reasons: the level of actual expenditures for a particular year may be affected by special events beyond the control of local governments such as earthquakes, floods etc.; and most importantly, the level of actual expenditures is influenced by quantity and/or the quality of services produced which can be above or below the standard level compatible with the standard expenditures. In order to make the comparison between standard and actual expenditure a good measure of efficiency. SOSE is developing a complementary methodology based on the evaluation of the level of services provided by each municipality through the estimation of a demand function which captures the relationship between context variables and output produced (see Porcelli et al., 2016 for more details about the evaluation of the standard level of services).

³ Given that 1 stands for the overall expenditure for a specific service, this allotment coefficient represents the share of expenditure attributable to each municipality. For example, if we refer to the education complementary services, Rome's allotment coefficient is equal to 0.086513706844, while Milan's one is equal to 0.050924900485. Thus, Rome's standard expenditure can be defined as 8.65% of the whole expenditure allocated for the education services that, in turn, is equal to 13.5% of the overall expenditure allocated for the fundamental municipal functions.

The rest of the paper is articulated as follows. Section 2 provides a short analysis of the international and methodological context that forms the background for the technical choices adopted by SOSE. Section 3 describes the main features of the Italian model, and section 4 shows how the model has been applied to the case of social care services, both sections are based on official documents produced by the Italian Ministry of Finance (MEF, 2014a 2014b). Section 5 discusses the evaluation of the standard level of services. Finally section 6 provides the main conclusions.

2. The international methodological context

2.1. Main institutional models

The list of possible methods for determining the expenditure needs of local governments is extremely varied. The level of heterogeneity is due not only to the variety of available techniques, but also to the manner in which they are applied by different countries which adopt variants and particular features based on their own historic and cultural traditions. Generally speaking, there are as many techniques as the number of countries that adopt fiscal equalization systems.

However, two essential dimensions emerge in all methodologies used to assess the expenditure needs of local governments when a top-down approach is followed as in the Italian case. The first defines the overall budget constraints (macro-budget), that is the global amount of transferable resources, and the second identifies the allotment mechanism adopted to allocate these resources among local governments.

This initial distinction is important, since the main purpose of the valuation of local governments' expenditure needs is to determine an allotment criteria for a predetermined amount of resources. In fact, standard expenditure needs are primarily a relative concept (relative needs), which only secondarily take on a validity in absolute monetary terms, particularly when used to finance basic levels of specific services.

Moreover, in spite of the extreme level of heterogeneity, the international literature (OECD, 1981, Blöchliger *et al.*, 2007, Reschovsky, 2007, Dafflon and Mischler, 2007) discerns two main methodologies to which it is possible to relate the practical experiences of different countries. The first based on actual expenditure needs (AEN), the second based on standard expenditure needs (SEN).

N.B: Copia ad uso personale. È vietata la riproduzione (totale o parziale) dell'opera con qualsiasi mezzo effettuata e la sua messa a disposizione di terzi, sia in forma gratuita sia a pagamento.

The two major approaches of AEN and SEN differ greatly in terms of information and calculation requirements. The AEN approach does not require the support of any statistical and/or econometric methodology, thus minimizing the need for data: expenditure needs are determined in relation to what was previously accounted for in the last available budget or in a certain number of past financial statements. However, though this approach has the unquestionable advantage of simplicity, it nonetheless presents some notable inconveniences such as elements of randomness in the distribution of equalization grants, the risk of perpetuating inequity in the distribution of grants over time, and the possibility of financial non-sustainability in the long run, as a result of soft budget constraints imposed on local governments.

For these reasons, the AEN approach is considered the least suitable system for the allotment of equalization grants over the long run. It can, however, play a useful role in the short term, when the transition towards more valid equalization formulas can require long and complex adjustment phases.

The SEN approach, though it may be adopted on the basis of different techniques, contrasts sharply from the AEN approach in that it aims to measure the expenditure needs of local authorities taking into account different characteristics of each local authority. The level of accuracy with which these expenditure differentials are taken into account depends on the quantity and quality of data used, as well as on the statistical tools adopted.

2.2. Standard expenditure needs

The criteria for SEN are based on the concept that the financial needs of a local authority are an expression of the territorial and socio-demographic characteristics of the resident population. These aspects, on the one hand, impact the needs of citizens and thus the demand for services; on the other hand, they directly influence production costs. In the case of municipal police services, for instance, a higher level of expenditure needs can be generated either by a greater number of vehicles, which increases the need for controls and the risk of accidents, or by an increased length of municipal roads, since it is more costly to patrol a vaster territory.

The advantage of overcoming the principle of AEN is linked to the allotment of equalization grants according to SEN, thereby combining equity and efficiency, two goals that are often difficult to achieve simultaneously. In terms of equity, the advantage is that of providing intergovernmental grants based on the real needs of each territory, guaranteeing that all authorities have enough resources to provide local services with uniform standards of

N.B: Copia ad uso personale. È vietata la riproduzione (totale o parziale) dell'opera con qualsiasi mezzo effettuata e la sua messa a disposizione di terzi, sia in forma gratuita sia a pagamento.

quality and quantity. From the efficiency point of view, the advantage is that of stimulating higher electoral accountability of local administrators, because expenditure level above standard expenditures cannot be covered by intergovernmental grants and must thus be financed directly by citizens through local taxes.

The OECD includes the adoption of the calculation of SEN within the best practices related to the planning of financing systems for local government. In particular, it is argued that the provision of equalization grants based on mathematical formulas that measure institutions' expenditure needs is preferable to systems based on actual expenditures or the discretion of the central government, since the mathematical formula approach guarantees greater transparency in the flow of grants, greater equity in the redistribution of resources, and greater efficiency in managing public expenditures, thanks to more rigid budget constraints for local governments.

The possibility of obtaining the concrete advantages attributed to the adoption of SEN criteria is strictly linked to the accuracy with which expenditure needs are calculated, as well as to the interpretability of results and the consequent possibility of transmitting them to the institutional entities concerned. For this reason, in order for the adoption of SEN criteria to prove effective, it is essential, first and foremost, to build broad ranging databases that provide the most accurate possible picture of the socio-economic context and territorial characteristics of each local authority. Secondly, advanced statistical and econometric techniques must be used to transform this information into expenditure needs as accurately as possible.

The calculation of SEN allows for the equitable and efficient provision of equalization grants, but also requires adequate technological support in order to work correctly. For this reason, it is employed mainly in economically developed countries, and rarely used by developing countries. Within the OECD, the method is used successfully in Denmark, Finland, Norway, Sweden, the United Kingdom, Holland, Portugal, Japan and Australia.

Among European countries, the United Kingdom is the one that has been using sophisticated statistical and econometric techniques to determine the SEN of local authorities the longest. After more than a decade of debate and experimentation, the introduction of standardization techniques for the valuation of expenditures needs of local governments became operational during the early '90s, together with a process of renewal across the local public sector. This program has placed the UK at the forefront of countries that are most attentive to issues related to efficiency and performance measurement in the delivery of local services.

N.B: Copia ad uso personale. È vietata la riproduzione (totale o parziale) dell'opera con qualsiasi mezzo effettuata e la sua messa a disposizione di terzi, sia in forma gratuita sia a pagamento.

In the broader global scenario, the fiscal equalization system adopted in Australia is considered one of the most complete and is used as a model for countries such as Italy which are undertaking reforms in this direction. One of the aspects of the Australian system that is generally appreciated is its choice to entrust the technical responsibility of determining standards to an independent agency, the Commonwealth Grants Commission.

2.3. The main techniques for determining standard expenditure needs

The methodologies for determining SEN are characterized, on the one hand, for the quantity of data used to capture the environmental and socioeconomic characteristics of the territories of the local authorities being analyzed, and on the other hand for the statistical techniques used to select and evaluate the weight these variables have in determining differentials expenditure needs.

The simplest and least refined standardization approach is that of *uniform per capita expenditure*, in which the size of the resident population is used as the sole determining factor for expenditure needs. Costs are standardized based on a uniform per capita amount for all authorities, equal to the macrobudget divided by the resident population: expenditure needs for each authority are obtained by multiplying this per capita value by the resident population.

A more complete approach to the calculation of expenditure needs, which includes other variables in addition to the size of resident population, is the *Representative Expenditure System* (RES). In this approach, the expenditure needs for each authority are determined by the linear combination of a series of load factors obtained in relation to the weight these factors have in determining expenditure needs. Load factors can include environmental characteristics such as surface area, number of residents, composition of the resident population by age, length of roads, etc., or structural elements related to the quantity of services produced and the inputs employed for supplying local public services. Usually, the choice of these factors is entrusted to experts or to decisions of a political nature, while weights are normally determined using statistical methods.

A very simple case of RES occurs when the SEN for a given region depend only on two factors with pre-established weights: the resident population and surface area. For example, given a region where 10% of the national population live and whose surface area amounts to 50% of the country's overall surface area, if the weights are 0.8 for the population and 0.2 for the

N.B: Copia ad uso personale. È vietata la riproduzione (totale o parziale) dell'opera con qualsiasi mezzo effettuata e la sua messa a disposizione di terzi, sia in forma gratuita sia a pagamento.

surface area, the region's SEN will be equal to $0.18 = 0.8 \times 0.1 + 0.2 \times 0.5$. The result is that the region will be entitled to 18% of all equalization grants (supposing the equalization is determined solely in relation to expenditure needs).

The RES approach can provide acceptable results when the authorities analyzed are few in number and economically and socially similar. However, its application becomes less effective when the mixture of local authorities has markedly varied characteristics. The immediacy and flexibility of the RES are not supported by a theoretical model that can act as a guide in the selection of load factors and weights, which, in essence, are discretionary.

For these reasons, the majority of developed countries that allot equalization grants based on SEN adopt an approach that is similar to the RES but more sophisticated in its application. This is the *Regression-based Cost Approach* (RCA), based on which the SEN of each authority are assessed as the expected value for a cost function (i.e. linked to the efficient behavior of the local government), estimated using multiple linear regression techniques.

With the RCA approach, the selection of variables to be inserted in the estimation model is guided by a theoretical framework based on the interaction between the demand for public services expressed by citizens and the supply of public services expressed by the local government (see Porcelli *et al.*, 2016 and also Barabaschi *et al.*, 2014) for more details about the theoretical framework). The result is that the efficient cost of supplying a given service depends on three essential groups of variables: the optimal quantity of service offered⁴; prices for the inputs used in the production process (primarily labour costs); and the context variables related to the supply side, i.e. external factors that, with other conditions being equal, can favor or hinder the supply of local public goods (e.g. the morphological characteristics of the territory, or the extension of its surface area).

The weight with which each variable affects the determination of SEN is estimated in a statistically robust manner through the use of multiple regression techniques. In this way all factors for calculating cost differentials are determined within a model that is capable of correctly representing the variables that identify the real determinants of expenditure needs, especially in the case of extremely heterogeneous local authorities.

Compared to the RES approach, the RCA approach provides greater accuracy, but also requires greater quantity and quality of data, particularly in

⁴ Optimal service quantity refers to services that best satisfy the preferences and/or needs of resident citizens.

Copyright © FrancoAngeli

the measurement of input prices and the correct identification of output variables that can be used to measure the optimal quantity of public services supplied in equilibrium.

In particular, the presence of public service output variables within the estimation model for SEN presents two orders of problems. Firstly, the outputs are not always measurable and there may be therefore a very serious lack of information. Secondly, even if the public service outputs are observable with extreme accuracy (e.g. in situations in which the state establishes basic levels of services to which individual local authorities must conform) they can be endogenous, since the supplied quantity is determined jointly with the level of expenditures. In this last case, to correctly estimate the weight of an output in determining expenditure needs, it would be necessary to employ a two-stage estimation techniques based on the use of instrumental variables that cannot always be adopted successfully.

A commonly used solution for resolving both problems reported above is to estimate a variant of the cost function, known as an *expenditure function*. In essence, an expenditure function is a cost function in which the optimal quantity of the service provided is substituted by its determinants, represented by the background variables that impact the demand of citizens (e.g. the composition of the population by age, since an older population will express a larger demand for social support services).

For this reason, in the end, an RCA methodology based on the estimate of an expenditure function is the most complete and robust approach for evaluating SEN, and it is therefore the approach that is commonly used globally. This variant to the pure RCA approach is very similar to the RES approach in appearance, with the essential difference that the selection of variables used for the calculation is supported by a theoretical model and that the weights of these variables are determined by a multiple regression analysis, which leaves little space for discretionary interventions, thereby rendering the estimate of expenditure needs more robust and objective.

Compared to the pure RCA approach, the RCA approach based on the estimate of an expenditure function has the advantage of not having to adopt any further econometric instruments, other than basic multiple regression, in correctly estimating the impact exercised by the quantity and quality of outputs in determining standard expenditure levels. On the other hand, due to this simplification, it does not allow for the measurement of the direct relationship between expenditure needs and outputs, and thus it does not allow the direct estimation of expenditure needs based on the level of optimal public services that should be supplied in equilibrium.

Copyright © FrancoAngeli

N.B: Copia ad uso personale. È vietata la riproduzione (totale o parziale) dell'opera con qualsiasi mezzo effettuata e la sua messa a disposizione di terzi, sia in forma gratuita sia a pagamento.

3. Determining standard expenditure needs in Italy

3.1. Expenditure functions vs Cost functions

The assessment methodology for SEN deemed most suitable for Italy was the RCA Approach, both in its pure form based on the estimation of a cost function and in its reduced form based on the estimation of an expenditure function (see SOSE, 2012, 2013). This method was preferred over the RES approach given the heterogeneity and complexity of municipalities. Moreover, the RCA approach, which is supported by a theoretical model that allows for the valuation of expenditures needs through the estimation of a cost function, fully embodies the legislative provisions that impose to valuate SEN, taking into account the degree of efficiency in the provision of local services.

However, faced with the extreme heterogeneity of the essential services being assessed, it is necessary to adapt the RCA to the specific characteristics of each function, adopting diverse estimation models. The discriminating element for the choice of a specific model is the characteristic of the variables, which can be used for measuring the output provided, i.e. the possibility of considering the quantity of services provided in calculating SEN.

The two main characteristics of the essential services implemented by local authorities are the level of measurability of the services provided and the extent to which these services are exogenous, with respect to the decisionmaking autonomy of each local authority.

Based on the combinations of these two output characteristics, three groups of services can be identified, each of which corresponds to a different model for the assessment of the expenditure needs. Table 1 below illustrates this ranking outline. (Barabaschi *et al.*, 2014) provide also a clear discussion about the differences between the procedure that uses an estimate of the standard cost and the one that estimates an expenditure function.

		Services are exogenous to the decision-making autonomy of each local authority			
Yes No					
Measurable	Yes	Pure RCA (cost function)	Expenditure function RCA or Pure RCA according to the possibility of testing the endogeneity of output variables		
Services	No		Expenditure function RCA		

The first group (north-west quadrant) includes cases for which the quantity of services provided can be measured satisfactorily and the output variables capturing all the essential factors of the implemented activity are exogenous to

1	3	3
1	-	-

Copyright © FrancoAngeli

the decision-making autonomy of each local authority. In these cases, we can proceed with estimating SEN using a pure RCA approach, effectively obtaining an estimation model of SEN based on the product between exogenous standard (or minimal) service levels and standard average costs.

The second group (north-east quadrant) includes cases for which the quantity of the service provided can be measured, but these services (in terms of quantity and/or quality) reflect the discretional choices of each local authority. In these cases, the estimation of SEN can be performed robustly and consistently using the RCA approach, based on the estimation of a cost function, which easily and effectively overcomes the problem of endogeneity for output variables with respect to actual current expenditure. Otherwise the estimation of a performed robustly approach based on the estimation of an expenditure function where output variables are replaced by exogenous demand factors.

The third group (south quadrant) includes those sectors for which it is not possible to measure the quantity of services provided since measurements of available outputs do not allow for a complete picture of the activities of the local authority. In these cases, the estimation of expenditure needs follows the RCA approach based on the estimation of an expenditure function. The majority of functions for municipalities fall in this group.

3.2. The data

The methodology for gathering data is one of the main contributions provided by the Italian experience to the international literature of SEN valuation. The collection of data through questionnaires has allowed SOSE to effectively resolve what is generally deemed to be the primary weak point of estimation mechanisms for SEN: lack of information. Larger volumes of data collected and processed for the determination of SEN ensures higher degree of robustness in the estimation process.

Though the questionnaires, which were prepared in collaboration with the Italian Association of Municipalities and distributed through a dedicated web portal, local authorities were requested to provide information relating to accounting and structural factors in the implementation of services for essential services.

The questionnaires respond to the need to integrate/reclassify and correct what is already available from official sources. In particular, the data contained in the Budget Sheets provided by the Ministry of Interior proved insufficient in determining the actual expenditures of services object of the

N.B: Copia ad uso personale. È vietata la riproduzione (totale o parziale) dell'opera con qualsiasi mezzo effettuata e la sua messa a disposizione di terzi, sia in forma gratuita sia a pagamento.

standardization analysis. Although the use of Budget Sheets as a starting point has the unquestionable advantage of providing greater stability for gathered information, the extreme heterogeneity of the criteria for recording accounting data and the summary nature of the document prevent Budget Sheets to be credible as the sole source for the determination of actual current expenditures used as dependent variables for the estimation model.

Moreover, other official sources also do not provide enough detailed information to analyze the main features of the services provided by local governments. For example, no official source exists to provide information neither about outputs nor about the main inputs employed in the production process.

The structural data collected through the questionnaire thus represents an extraordinary innovation, allowing elements such as inputs (personnel employed, capital goods, local units used, etc.), outputs (services implemented), and procedures for the implementation of services (unions of municipalities and other forms of joint or direct provision) to be assembled into a unitary framework. On the other hand, accounting information requested in the questionnaires allow for a reclassification and integration of the data contained in the Budget Sheets, thus providing a more accurate representation of actual expenditures.

Therefore, prior to undertaking the valuation of SEN, a unique database was built, including accounting information from the archives of the Budget Sheets, provided by the Ministry of the Interior for the year 2009 and 2010, structural information collected with the questionnaires and information related to socio-economic context derived from official sources. The database provides information about 6702 municipalities and 220 unions⁵ of municipalities. In the process of data collection, the methodological contribution provided by ISTAT (National Statistic Institute) proved essential. A detailed analysis of the determinants of municipal standard expenditure needs for the Italian case is reported by Brunello *et al.*, 2015.

SOSE also put great effort into controlling the quality of data. For example for the questionnaire FC02U related to municipal police services, direct contacts were established with 486 local authorities via ordinary and certified electronic mail, as well as by telephone. This activity allowed for the correction of 1076 irregularities, corresponding to cases of serious incoherence in the declared data.

In addition, the information acquired through the questionnaires made it possible, for the first time ever, to assess the organizational models used by local authorities for the provision of services, thereby allowing the valuation

⁵ Maximum number of unions registered in the provision of local police services.

Copyright © FrancoAngeli

of their impact on expenditure needs, as stipulated under Article 4 paragraph 3 of Legislative Decree 216/10. In particular, for the first time it has been possible to accurately assess the geography of unions of municipalities and other forms of joint provision of local public services (mountain communities, consortiums and agreements), of which very little was known before.

3.3. The theoretical framework of standard expenditure needs

The theoretical framework followed by SOSE is based on the interaction between the supply and demand of local public services, both expressed in per capita terms with respect to the number of beneficiaries designated here as the client group (see Porcelli *et al.*, 2016 and also Barabaschi *et al.*, 2014 for more details about the theoretical framework). In the majority of cases the client group corresponds to the total resident population; in the cases of education complementary services and nursery services, however, the client group corresponds to children aged 3-14 and aged 0-2, respectively.

The demand for local public services (g_e) in the equation (1) depends on: the background variables related to the demand for local public services (**Q**), which capture the demographic and socio-economic aspects influencing the demand; the average income (**R**); and the per capita cost (with respect to the client group) of local public services (y):

$$g_e = d(\boldsymbol{Q}, \boldsymbol{R}, \boldsymbol{y}) \tag{1}$$

The supply for local public services (*y*), expressed in terms of per capita costs, is reported in the equation (2) and depends on the following variables: the level of exogenous load factors $(g_s)^6$; the level of endogenous outputs for public services (g_e) ; input prices (**p**); and background variables related to the supply (**A**)⁷:

$$y = s(g_s, g_e, \boldsymbol{p}, \boldsymbol{A}) \tag{2}$$

⁶ These are, for instance, services provided by local authorities for higher-level administrations (such as crime notifications, lawsuits and disputes received or the notification of traffic accidents, in the case of local police functions of municipalities) and, generally, activities not directly attributable to the decisions of local administrators.

⁷ These are, for instance, morphological and socio-economic factors that do not affect preferences regarding the level of public services (i.e., those exogenous factors that can favor or hinder the supply of local public goods, such as economies of scale measured through the size of resident population.

136

Copyright © FrancoAngeli

The optimal level of local public services and their costs are thus determined simultaneously within a structural model that includes two equations: (1) and (2).

In some cases the estimation of SEN can be performed directly using the cost function reported in the equation (2): that is a model that multiplies the quantities of the service provided by coefficients expressing the standard average cost of individual outputs comprising the services implemented (pure RCA). This option can be suitably considered in certain specific cases. The first is when local public services can be measured adequately, and are represented by exogenous load factors (e.g. in the case of complementary education services provided by municipalities). The second is when local public services are represented by endogenous outputs that can be measured with an excellent degree of accuracy and the problem of output endogeneity can be tested through the identification of a set of valid instrumental variables. In the case of nursery services provided by municipalities, for example, the endogeneity of the output variable (the number of children attending fulltime and part-time) has been tested using the following instrumental variables: the rate of female employment (age 30-54) registered at municipal level in 2001, and the rate of commuters entering in each municipality in proportion to the resident population.

Alternatively, in order to combine simplicity and robustness of estimates, the valuation of expenditure needs can be implemented using an expenditure function, which represents a reduced form of the structural model for supply and demand for public services. The expenditure function used to estimate SEN is reported in equation (3), which is obtained by substituting equation (1) into equation (2), thus expressing the relationship between the costs for the provision of local services and the local context in which the local government operates.

$$y = f(Q, R, \boldsymbol{p}, \boldsymbol{A}, \boldsymbol{g}_s) \tag{3}$$

The primary advantage of evaluating SEN through the expenditure function is that consistent estimates can be obtained easily and robustly using the *Ordinary Least Squares* (OLS) estimator, since the independent variables are represented only by exogenous variables. This is the main reason why the estimation of expenditure needs based on expenditure functions is the most commonly adopted econometric approach at a global level, while the pure RCA approach is rarely adopted.

Copyright © FrancoAngeli

3.4. The empirical model of standard expenditure needs

Regardless of the theoretical model selected (cost function or expenditure function), the estimation of SEN is implemented through the calculation of the expected values from an empirical model that identifies the relationship between the current per capita actual expenditures (dependent variable) and a set of independent variables within a multivariate regression model.⁸

Usually the following groups of independent variables have been used:

- Output variables (specific to the cost function approach), measuring the quantity and quality of the services provided (e.g. the number of children using nursery services, the number of pupils using meal services etc.);
- Demand background variables (specific for the expenditure function approach), measuring the demographic and socio-economic aspects required to capture local preferences/needs regarding the demand for public services (e.g. the percentage of elderly people on total population, the number of people with addictions and mental health problems, the number of single-parent families etc.);
- Supply background variables, comprising environmental characteristics that impact the total productivity of input factors (e.g. the numerousness of the resident population, which captures congestion phenomena and/or economies of scale, the morphological characteristics of the territory etc.);
- Exogenous load factors, capturing the impact of services provided by local authorities on behalf of higher level administrations, or of activities not directly attributable to the decisions of local administrators (e.g. crime notifications, lawsuits and disputes received, or the notification of traffic accidents in the case of local police services; the front-office activity related to the issue of vital certificates carried out as part of registry office services etc.);
- Input prices, represented in the majority of cases by the average staff expenditures per employee computed as the ratio between the total labour costs and the total equivalent number of employees reported in the ques-

⁸ In the construction of the empirical model of standard expenditure needs, current expenditures are usually considered in per capita terms and, therefore, have been divided by the resident population, which constitutes the client group for the majority of services. Moreover, the use of current per capita expenditures as a dependent variable is preferable, since it allows for a reduction of the heteroscedasticity problem in the data.

Copyright © FrancoAngeli

tionnaire, and the *average rents per square meter for commercial use* provided by the *Osservatorio del Mercato Immobiliare* (OMI)⁹ of the Territorial Agency¹⁰;

- Managerial choices, represented by a set of variables used to capture the impact exerted on expenditure by organizational choices such as the joint provision of local public services with other authorities (unions of municipalities, conventions etc.) or the staff/child ratio (in the case of nursery service);
- *Regional fixed effects*, representing a set of dummy variables included in the model in order to capture the impact of regional polices.

As a result the estimation of SEN is conducted using the following empirical model:

$$y_i = \alpha + \beta' X_i + \gamma' W_i + \delta' Z_i + \varepsilon_i$$
(4)

where:

- *i* corresponds to the local authority index;
- α , β , γ , δ are the coefficients to be estimated;
- y_i is the dependent variable corresponding to the current per capita expenditures;
- \mathbf{X}_i is the vector of independent variables which are used both in the estimation and in the subsequent stage of calculating SEN. As evident in equation (3), these are demand background variables (Q), supply background variables (A), and variables related to exogenous load factors (g_s).
- \mathbf{W}_i is the vector of independent variables which, while used in the estimation, become "*target*" variables when computing the SEN. These variables should be used as stated by each local authority in estimating the empirical model, so as to avoid distortions in the coefficients; however, in computing the fitted values, this variable can assume a target value (W^{*}) to be attained by the local authority. The W_i variables include the input prices (*p*) and endogenous outputs (*g_e*) when using the cost function approach.
- \mathbf{Z}_i is the vector of independent variables that can explain cost differences related to organizational models, particular managerial choices, regional fixed effects and the average municipal income. These variables are included in the empirical model to avoid the problem of omitted variables, however, in computing SEN the impact of these variables is ignored or

⁹ Real Estate Market Centre.

¹⁰ The average rents per square meter for office use is a proxy of the price index of other productive factors different from labour.

Copyright © FrancoAngeli

"neutralized", in order to attribute to each municipality a standard expenditures need that does not depend on specific organizational choices adopted, form of service management, and the impact of regional policies.

- ε_i represents the idiosyncratic error term, with zero mean, uncorrelated with \mathbf{X}_i , \mathbf{W}_i and \mathbf{Z}_i , but potentially heterosckedastic.

The specification of the empirical model has been conducted using the "stepwise" method validated by a "general-to-specific" and "specific-to-general" approach. Once the regressors have been selected, the empirical model is estimated using an OLS estimator with robust standard errors in order to control for the heteroscedasticity in the covariance matrix (the only exception is represented by nursery services)¹¹.

3.5. The computation of standard expenditure needs

Once the coefficients of the empirical model of SEN reported in equation (4) have been estimated, the expected values (\hat{y}_i) of the current expenditure of each municipality (also considering those excluded from the regression sample) are obtained as follows:

$$\hat{y}_i = \hat{\alpha} + \hat{\beta}' X_i + \hat{\gamma}' W_i + \hat{\delta}' Z_i$$
(5)

Subsequently, the *Theoretical Standard Expenditure Needs* (FS) are computed, neutralizing the impact of variables Z_i and replacing the W_i variables with target values (if available) as reported in equation (6):

$$FS_i = \hat{\alpha} + \hat{\beta}' X_i + \hat{\gamma} W^* + \hat{\delta}' Z^*$$
(6)

It follows that the difference Δ_i between the value of actual current expenditure and the theoretical standard expenditure needs is equal to:

$$\Delta_i = y_i - FS_i = \hat{\gamma}(\boldsymbol{W}_i - \boldsymbol{W}^*) + \hat{\delta}'(\boldsymbol{Z}_i - \boldsymbol{Z}^*) + \hat{\varepsilon}_i$$
(7)

¹¹ The following statistical tests have been also used to ensure the robustness of the estimates: Cook's distance (Cook D), commonly used to estimate the impact of a single observation on OLS coefficient estimates; the leverage points analysis, which identifies the observations most distant from the corresponding fitted values; the coherence of "studentized" residuals (Student R); and the analysis of dfbetas, which examine the observations that significantly influence parameter estimates. In the end, observations identified as outliers or extreme values based on the statistical tests have been eliminated by the regression sample.

Copyright © FrancoAngeli

The difference Δ_i can be decomposed into three additive components, where:

 $\Delta_i^1 = \hat{\gamma}(\boldsymbol{W}_i - \boldsymbol{W}^*)$ represents the difference corresponding to the impact exerted by target variables; $\Delta_i^2 = \hat{\delta}'(\boldsymbol{Z}_i - \boldsymbol{Z}^*)$ measures the variability connected to particular managerial choices, and regional fixed effects; $\Delta_i^3 = \hat{\varepsilon}_i$ represents the residual part of the difference between actual expenditures and standard expenditures due to the stochastic component of the model.

After estimating the FS it is possible to calculate an indicator of relative needs for each municipality, corresponding to the allotment coefficient FS_CR_i reported below in equation (8):

$$FS_CR_i = \frac{N_i \cdot FS_i}{\sum_i N_i \cdot FS_i}$$
(8)

where N_i is the size of the client group of each municipality, which corresponds to the total resident population in the majority of services (in the case of education complementary services and nursery services a slightly different procedure has been followed because of the logarithmic specification of the empirical model of the cost function).

The FS_CR_i represents the final result of the procedure, since the valuation of SEN leads to the computation of a set of allotment coefficients for an overall expenditure amount determined externally by the central government. The allotment coefficients computed for each function can then be combined to calculate an overall allotment coefficient for the total amount of essential services' expenditures.

Note that, in the case of unions of municipalities, the estimated SEN were calculated on a "consolidated" base, with reference to the union (with few exceptions such as variables related to the economies of scale) and then distributed to each municipality belonging to the union proportionally to the resident population of each municipality.

Table 2 provides a general overview of the final results reached in the estimation of municipal standard expenditure needs (SEN), reporting the main technical issues for each of the 12 services.

In the majority of cases the actual expenditure adopted as dependent variable refers to 2010 values, only for local police and general administrative services the reference year is 2009. Although it would be preferable to use

Copyright © FrancoAngeli

the actual expenditure from the same year as dependent variable for all services, it must be highlighted that, since the valuation of SEN seeks to compute an allotment coefficient, the choice of reference year for the actual expenditure exerts a negligible impact on the final results given the stability in the distribution of the current expenditure across sectors.

With regard the client group, it can be noted that the total resident population has been chosen in the majority of cases. The only exceptions are represented by education complementary services and by nursery services, where the client group is represented respectively by children aged 3-14 and aged 0-2.

Functions and services		Questionnaire	Actual expenditure	Client group	General approach	Model specification	Estimator	Dependent variable	R-sq.	Regression sample (n.)	Unions of municipalities (n.)
	Tax office	FC01A							0.4425	3991	152
General	Technical office	FC01B		Total	Evpondituro			Current actual	0.4187	3472	157
Management and Control Functions	Civil registry	FC01C	2009	resident population	function	Linear	OLS	per capita	0.6029	4761	137
	General services	FC01D						expenditure	0.7322	3882	178
Local Police		FC02U	2009	Total resident population	Expenditure function	Linear	OLS	Current actual per capita expenditure	0.3726	5061	220
Education (complementary services)		FC03U	2010	Population aged 3-14	Cost function	Logarithmic	OLS	Log of total current actual expenditure	0.9254	3990	183
Public	Public roads			Total				Current	0.6484	5333	157
Roads and Transport	Local public transportation	FC04U	2010	resident population	Expenditure function	Linear	OLS	actual per capita expenditure	0.5829	915	157
Planning and Environmental Functions	Land management and planning	FC05U	C05U 2010	Total resident population	al Expenditure	e Linear	ar OLS	Current actual	0.2773	3163	157
	Waste management				Idifotori			expenditure	0.6462	3989	157
Social Care	General social services	ECOSI	2010	Total resident population	Expenditure function	Linear	OLS	Current actual per capita expenditure	0.3923	4603	175
	Nursery services	10000	2010	Population aged 0-2	Cost function	Logarithmic	LAD (quantile regres- sion)	Log of total current actual expenditure	0.6783	1133	175

Tab. 2 – Overview of some technical issues regarding the estimation of municipal standard expenditure needs

Copyright © FrancoAngeli

When considering the choice of the general theoretical approach, of the empirical model specification and of the estimator employed in the analysis, it is possible to divide the whole set of twelve services into two groups: one including 10 services related to the general administrative sector, the local police, public road and transport, planning and environment and general social services, for which an expenditure function approach has been followed using a liner specification of the empirical model and OLS estimator; and a second including education complementary services and nursery services, for which a cost function approach (pure RCA) has been followed, estimating a logarithmic cost function through OLS estimator for the former service and LAD estimator for the latter.

It is important to emphasize, in conclusion, that the choice of using a logarithmic specification for the cost functions and a linear specification for the expenditure functions was not determined by the different theoretical model adopted as a reference point for the empirical model. This choice was driven, instead, by the necessity to consider the specification which provided the best fit in order to minimize the distance between historical and standard expenditure as requested explicitly by the association of municipalities. Similarly the choice to used LAD estimator rather than OLS was driven by the necessity to maximize the fit of the model as well as to avoid technical problems in generating the expected values.

4. The case of the social care services

As an example, we discuss in more details the main methodological aspects regarding the estimation of the SEN of the social care sector, which represents one of the main function performed by municipalities and absorbs about 20% of the total actual current expenditure, corresponding to more than 7 billion euros (2010 figures)¹². A detailed analysis about the model used to estimate SEN for the Register Office, instead, is provided by Barabaschi *et al.*, 2014.

Information derived from official sources has been supplemented with accounting and structural data collected through questionnaire FC06U with reference to the year 2010. The questionnaire, similarly to those imple-

Copyright © FrancoAngeli

¹² Although most of the data are publicly available through *www.opencivitas.it*, the reproducibility of the estimates is not immediate because, for privacy reasons, some variables can be obtained only after a specific request addressed to the Ministry of Finance.

mented for other functions, consists of twelve tables, each with different objectives and content. The top six tables contain mainly structural data about managerial choices, staff structure, other inputs employed in the production, and the quality and quantity of services. The remaining six tables are mainly used to collect accounting information aimed at supplementing the information derived from the Budget Sheets in order to evaluate the level of actual expenditures used as dependent variable in the empirical model. In particular, they provide information about the structure of staff cost, the outsourcing, and revenues from the provision of services. Regarding nursery services, for example, questionnaire FC06U provides a wide range of new information about staff structure, the number of children attending part-time and full-time, the number of voucher recipients, opening days and opening hours, the percentage of children using the meal service etc.

In sum, 6,667 municipalities (99.5% of the total) returned the questionnaire within the established deadline; subsequently, SOSE put great effort into controlling the quality of data by establishing direct contacts with 3,982 local authorities via ordinary and certified electronic mail, as well as by telephone. This activity allowed for the correction of 8,107 irregularities, corresponding to cases of serious incoherence in the declared data.

Based on the data collected, the services provided in the social care sector have been divided into 11 macro-services (or macro-output):

- nine are potentially endogenous with respect to the choices of local administrators: nursery services; early childhood (e.g. other educational and recreational services for early childhood); social emergency (e.g. meals provided by the social canteen); social inclusion (e.g. transport of people with disabilities, literacy courses); income support (e.g. rent allowances); home care (e.g. home care meal service, tele-assistance, remote monitoring); day care centers (e.g. recreational social and cultural activities); residential care (e.g. children in foster care, assistance and social rehabilitation); back office activity;
- two can be classified as exogenous load factors: front office services; cemetery, cremation and mortuary services.

Each macro-output (excluding nursery services) is, in turn, divided into several basic services for a total of 36 elementary output variables.

In consideration of the specific nature of each macro-service, the estimation of SEN has been carried out by separately analyzing the nursery services, which have children aged 0-2 as client group, and by grouping the rest of the social services which have the entire resident population as client group within a single category called general social services.

Copyright © FrancoAngeli

N.B: Copia ad uso personale. È vietata la riproduzione (totale o parziale) dell'opera con qualsiasi mezzo effettuata e la sua messa a disposizione di terzi, sia in forma gratuita sia a pagamento.

*4.1. The nursery services*¹³

With regard to nursery services, the output produced can be easily measured through the number of served children. Preliminary tests conducted to verify output endogeneity did not reject the null hypothesis according to which the amount of children served can be considered exogenous with respect to municipal expenditure. Consequently it was decided to evaluate the SEN through the estimation of a cost function using a log-log specification of the empirical model, as reported in the following equation (9):

$$log(Y_i) = \alpha_0 + \gamma log(M_i) + \theta_1 log(P_{1i}) + \theta_2 log(P_{2i}) + \delta'_2 F_i + \delta'_3 L_i + \delta'_{1e} S_i e_i + \delta'_{1(1-e)} S_i (1-e_i) + \varphi_1 D_i + \varphi_2 D_i log(V_i) + \varepsilon_i$$
(9)

where: *i* is the municipal index, α_0 is the constant term, *Y* corresponds to the actual current expenditure, M is the output variable measured in terms of served children; P_1 is the variable capturing the price of labour (computed as the weighted average between internal and external labour costs), P_2 is the variable capturing the cost of capital (proxied by average rents per square meter for commercial use), *F* is the vector of cost-shift variables related to fixed cost components and/or the service quality (total area in square meters and staff/child ratio), *L* is the vector of binary variables related to the forms of joint provision and regional fixed effects; *S* is a vector of cost-shift variables that measure the intensity of service¹⁴; *e* represents the share of children which attend nursery not run directly by the municipality because externalized to private entities; *D* is a binary variable that takes value 1 if the municipality provides only vouchers; *V* corresponds to the number of voucher recipients, ε is the idiosyncratic error term of zero mean, uncorrelated with the other independent variables and potentially heteroscedastic.

The analysis of managerial choices provides the following segmentation of local authorities: 1,783 municipalities directly provide nursery services; 552 municipalities provide the service jointly with other authorities (unions of municipalities, mountain communities, conventions etc.); 4,367 municipalities do not provide nursery service.

¹⁴ This set of variables includes: the share of part-time children, the share of children who use meal services, the share of infants, the dummy that identifies opening hours of fewer than seven hours, the dummy that identifies opening days of fewer than 213 days and the variables related to the presence of kitchens.

Copyright © FrancoAngeli

¹³ This section is based on official documents produced by the Italian Ministry of Finance (MEF 2014b).

The sample used for the estimation of the cost function has been defined after an detailed analysis of the data reported only by those municipalities that provide nursery services. In this way we have constructed a sample of 1,133 local authorities (including 12 unions, 4 mountain communities, 6 consortia, and 33 conventions). After the removal of outliers, the regression sample shrinks to 970 observations.

Despite the elimination of outliers from the regression sample, OLS estimates showed a poor fit, as the value of the *root mean squared error* was very high making the transformation of the expected values of the logarithmic function into monetary values - a necessary transformation for the correct valuation of the SEN - impossible. To solve this problem, it was decided to compute the coefficient point estimates of the parameters of equation (9) using the *Least Absolute Deviations* (LAD) estimator¹⁵, which corresponds to a quantile regression analysis performed in relation to median values. The estimates of the coefficients of the independent variables in all other percentiles have also been taken into consideration, however, in order to verify the robustness of the point estimates obtained at the median, thus showing a good stability of the coefficients' point estimates of all independent variables.

The coefficients' point estimates of the cost function are reported in Table 3a in relation to different groups of variables: X, W, and Z. Tables 3b and 3c paper the values considered for the computation of SEN regarding the variables included in group W and group Z. Final estimates show a pseudo R-squared value equal to 0.6783 providing evidence of a good fit for the empirical model. It is possible to note, in particular, the presence of slight increasing economies of scale, since the coefficient of output is below one. It should also be noted, moreover, that the null hypothesis of price elasticity equal to one cannot be rejected.

The logarithmic specification of the final model allows us to interpret the coefficient estimates in terms of elasticity; however it makes an immediate valuation of the theoretical SEN (FS) impossible. In this case we need to use the following formula to evaluate the FS for each municipality *i*:

$$FS_i = \frac{\exp(\hat{\alpha} + \hat{\beta}' X_i + \hat{\gamma}' W_i^* + \hat{\delta}' Z_i^*)}{N_i}$$
(10)

where exp(.) is the exponential function. As a results, individual components of the *FS* cannot be considered independently from the others since, in this case, the *FS* does not have an additively structure.

¹⁵ For more details about the LAD estimator see Cameron and Trivedi (2010).

146

Copyright © FrancoAngeli

The absence of basic levels of services established by law according to which a minimum percentage of children aged 0-2 must be served by nursery services, prevents SOSE from listing the output variables in the set of target variables W. This implies that the *FS* of each local authority can be computed considering only the actual number of children served by nursery services, papering zero SEN for those municipalities without nursery services¹⁶. The valuation of a positive standard expenditure also for those municipalities without nursery service is subordinate to the identification of the some mandatory "essential level of services" by the policymakers.

	.1004					
X variables						
OUTPUT Number of children attending full-time and part-time (log)	.9412	***				
Share of infants (part-time and full-time)	.2783	**				
IN LENSILY OF SERVICE Share of children using meal services (.2305	**				
Share of children attending part-time sessions -0	.2449	**				
Share of children in kindergartens not directly managed by the local government	.5397	***				
MANAGEMENT Presence of kitchens (dummy, 1 = yes) X share of children under direct management	.1461	**				
CHOICES Presence of kitchens (dummy, 1 = yes) X share of children not under direct management	.3350	***				
Days open fewer than 213 days (dummy, 1 = yes) -0	.0858	**				
Hours open fewer than 7 hours (dummy, 1 = yes) -	.1856	**				
QUALITY Staff/child ratio	,8647	***				
OF SERVICE Total area in square meters (log)	.0918	***				
Vouches recipients (dummy, 1 = only vouchers) -0	.6238	**				
Number of voucher recipients (log) X Voucher recipients dummy	.7246	***				
W variables						
INPUT PRICES Average staff expenditures per employee (weighted average between internal and external labour costs) (log)	.5564	***				
Average rent per square meter for commercial use (log)	.3449	***				
Z variables						
ASSOCIATED Unions of municipalities (dummy, 1 = yes)	.2008					
MANAGEMENT Conventions of municipalities (dummy, 1 = yes)	.2607	**				
REGIONAL FIXED EFFECTS						

Tab. 3a - Nursery services, LAD coefficients' point estimates (coefficients can be interpreted as elasticities, dependent variable = log(actual current expenditure))

Bootstrap standard errors (200 repetitions), *** P-value < 0.001, ** P-value < 0.05, * P-value < 0.1

¹⁶ The presence of nursery services is identified by the positive number of served children (i.e., the number of children attending full time or part time, including the number of voucher recipients).

Copyright © FrancoAngeli

Tab. 3b – Nurserv services.	W^*	variables used i	n the	valuation of	^r standard e	xpenditure needs
						T

Average staff expenditures per employee (average between internal and external labour costs)	Internal component, as reported in the questionnaire. External component, replaced with the median value computed by the region and by population brackets.
Average rents per square meter for commercial use	Median value by the region and the population brackets

Tab. 3c – Nursery services, \mathbf{Z}^* variables used in the valuation of standard expenditure needs

Unions of municipalities	0
Conventions	0
Regional fixed effects	0

4.2. The remaining general social services¹⁷

With regard to the remaining general social services, the output presents a more complex structure. Furthermore, the difficulties encountered in testing the endogeneity of the output variables with respect to municipal expenditure, and the multi-output structure of the cost function, made the estimation of a cost function unfeasible at least at this stage. Consequently, the valuation of the SEN has been obtained using the expenditure function approach, specifying a linear empirical model using as a dependent variable the current expenditure in per capita terms with respect to the total resident population.

The analysis of managerial choices provides the following segmentation of local authorities: 6,124 municipalities directly provide general social services; 531 municipalities provide these services jointly with other authorities constituting a union; instead, only 12 municipalities do not provide any of the general social services.

The sample used for the estimation of the expenditure function has been defined after a detailed analysis of only those data reported by municipalities that provide at least one of the general social services. In this way we have constructed a sample of 4,603 local authorities including 37 unions of municipalities. After the removal of outliers, the regression sample shrinks to 4,400 observations.

¹⁷ This section is based on official documents produced by the Italian Ministry of Finance (MEF 2014a).

Copyright © FrancoAngeli

The coefficient point estimates of the expenditure function have been obtained using robust OLS estimator, and final results are reported in Table 4a in relation to different groups of variables: X, W, and Z. Tables 4b and 4c paper the values considered for the computation of SEN regarding the variables included in group W and group Z. Final estimates show a R-squared value equal to 0.3726 providing evidence of a good fit.

The linear specification of the expenditure function and the dependent variable expressed in terms of per capita actual expenditure allows us to interpret the coefficient point estimates in euros that multiplied by the related variables generate the additive components of the theoretical standard expenditure needs (FS).

Let us consider, for example, a hypothetical municipality of 25,000 inhabitants with all background variables and input prices equal to the sample mean. Considering that in the empirical model all background variables and input prices are expressed in deviation from the sample mean¹⁸, the FS of this hypothetical municipality can be computed as follows:

FS =	57.2419	+ (constant term)
	0.00096*25000	+ (impact of resident population)
	4.66401	+ (whether support services are provided)
	6.04295	+ (whether home health care services are provided)
	2.89174	(whether social integration and prevention ser- + vices are provided)
	7.08344	+ (whether residential care services are provided)

¹⁸ Note that this choice affects only the estimate of the constant term coefficient, since the other coefficient's point estimates are invariant to linear combinations of the regressors.

Copyright © FrancoAngeli

CONSTANT		57.2419	***			
	X variables					
	Female employment rate (per capita)	144.152	***			
	Foreign resident population (%)	0.32541	**			
	Elderly resident population (% over 65)	0.58844	**			
	Mortality rate (average 2008-2010)	0.37831	*			
BACKGROUND VARIABLES	Pupils with disabilities (pre-school, primary and second- ary; per capita)	1016.44	**			
(all variables entered in deviation from the sample	People with addictions and mental health problems (per capita)	587.912	***			
mean)	Number of crimes at provincial level (per capita)	119.157	**			
	Percentage of households with severe material depriva- tion (%)	1.14076	**			
	Number of single-parent families (per capita)	217.269	**			
	Number of survivor pensions (per capita)	133.698	**			
	Disability allowances (per capita)	315.613	**			
	Life expectancy without disabilities at age 65 (years)	-2.68469	**			
MANAGEMENT CHOICES	Support services (dummy, 1 = yes)	4.66401	**			
	Home health care (dummy, 1 = yes)	6.04295	**			
	Social integration and prevention services (dummy, 1 = yes)	2.89174	**			
	Residential care (dummy, 1 = yes)	7.08344	***			
	Population spline (3.000 - 10.000 inhabitants)	0.00229	***			
DISECONOMIES OF SCALE	Population spline (10.000 - 50.000 inhabitants)	0.00096	***			
	Population spline (100.000 - 110.000 inhabitants)	0.00349	***			
	W variables					
INPLIT PRICES (% deviation	Average rent per square meter for commercial use	0.05490	**			
from the sample mean)	Average staff expenditures per employee (weighted aver- age between internal and external labour costs)	0.06993	**			
Z variables						
INCOME (% deviation from the sample mean)	Income reported as a tax base for the national personal in- come tax	0.1532	**			
MANAGEMENT CHOICES	Reception for appointment (dummy, 1 = yes)	4.8058	***			
ASSOCIATED MANAGEMENT	Unions of municipalities (dummy, 1 = yes)	5.5299				
REGIONAL FIXED EFFECTS						

Tab. 4a – General social services, OLS coefficients' point estimates (all coefficients are expressed in euros, dependent variable = actual current expenditure/total resident population)

Robust standard errors, *** P-value < 0.001, ** P-value < 0.05, * P-value < 0.10

Copyright © FrancoAngeli

Tab. 4b – General social services, W^* variables used in the valuation of standard expenditure needs

Average staff expenditures per employee (average between internal and external labour costs)	Internal component, as reported in the question- naire External component, median value by the region and the population brackets
Average rents per square meter for commercial use	Median value by the region and the population brackets

Tab. 4c – General social services, \mathbf{Z}^* variables used in the valuation of standard expenditure needs

Unions of municipalities	0
Reception for appointment	0
Income reported as tax base for the national personal income tax	National average = 20.240,36 (euro, year 2010)
Regional fixed effects	0

Therefore, if all services listed in the set of managerial choices variables were provided, the *FS* would be equal to 101.92 euros. As a result, the *FS* of an identical municipality with a percentage of households with severe material deprivation 10% above the sample mean would be equal to euro 113.33 = 101.92 +1.14076*10, instead the *FS* of an identical municipality with a percentage of households with severe material deprivation 10% below the sample mean would be equal to euro 90.51 = 101.92 - 1.14076*10. Other examples can be provided along the same lines.

In conclusion, it is important to point out that although the FS can be computed for each local authority independently from the others, the SEN of each municipality, corresponding to the allotment coefficient FS_CR_i , are strictly dependent on the FSs of all other local authorities as reported in equation (8).

5. The model of standard levels of services

In addition to standard expenditure needs, the Italian law (Legislative Decree 216/10) prescribes also the evaluation of the level of services (output) provided by each local authority. The fact that the law requests that the expenditure needs of local government be measured, along with the quantity of services that should be offered, must be greeted favorably, since the latter is an extremely important exercise, both in rendering public expenditures efficient and in determining essential levels for services that local authorities

Copyright © FrancoAngeli

should provide in relation to their expenditure needs. This is a necessity which has been much debated politically and academically, but never clearly satisfied from a methodological standpoint.

However, unlike the evaluation of SEN, the measurement of the level of services is a quite new exercise and very few references can be found in the literature. Regarding the measurement of the historical level of output the English experience is, at the moment, the most interesting one since over the last twenty years the outputs of English local authorities have been measured systematically by the Audit Commission (see Audit Commission, 2009) through a range of indicators (over 200 active in 2009) known as BVPI¹⁹ and then used as one of the building blocks of the Comprehensive Performance Assessment (CPA) a system based on a balanced scorecard methodology which led to the annual publication of a ranking of local authorities, for eight years from 2002 to 2009, assigning to each authority a number of stars ranging from four for performance excellence, to zero for a highly negative judgment (see Lockwood and Porcelli, 2013 for more details about CPA).

The literature, instead, provides no help about the evaluation of the standard level of service (SLS) that should be used as a reference point to judge the level of output provided by each municipality. The methodology developed by SOSE suggests to interpret the SLS as the level of services (in terms of quantity and quality) compatible with the demand for services expressed by the resident population of a specific territory (for more details about the evaluation of SLS see Porcelli *et al.*, 2016). Therefore, starting from the same theoretical framework used for the evaluation of expenditure needs, the assessment of the standard level of services provided by each local authority is based on estimating the reduced form of the demand for local services reported in the equation (11), that is obtained by replacing the equation (2) in the equation (1).

$$g_e = h(Q, R, \boldsymbol{p}, \boldsymbol{A}, g_s) \tag{11}$$

When technically possible, the multi-output structure of services must be taken into account, and to correctly valorize any existing interdependencies between outputs it is important to estimate a structural model. In this context, the SLS of each municipality corresponds to the expected level of output $\widehat{g_{e}}$.

¹⁹ For example, among the output indicators for social care services are the number of elderly who require assistance at home in every 1000 inhabitants over 65 years of age (BPVI54); indicators related to environmental services include the percentage of recycled domestic waste (BPVI82a); and, lastly, indicators measuring the performance of general administrative services include the percentage of invoices paid within 30 days of receipt or within the agreed payment terms (BVPI8).

Copyright © FrancoAngeli

The joint analysis of the expenditure gap between historical expenditures and SEN, and the output gap between historical outputs (g_e) and SLS $(\widehat{g_e})$ can be used to evaluate the performance of each local government mapping municipalities into the four quadrants as shown if figure 1. The rationale behind the positioning of local authorities in the four quadrants is based on the consideration that SLS provide a measure of the potential demand corresponding to standard expenditures.

Therefore, a red light comes on for the local authorities in quadrants III and II of figure 1: local authorities positioned in quadrant III are designated "under standard" since they present actual expenditures that are lower than the standard expenditures and should satisfy a potential demand that is higher than the current supply capacity; local authorities positioned in quadrant II are designated "non-efficient" since in addition to facing a potential demand that is higher than the current supply capacity, they present actual expenditures that are higher than standard expenditures. Both these groups of local authorities should be placed under observation: the "under standard" authorities in order to ascertain that by obtaining more financial resources they effectively provide also more services in terms of quantity and/or quality, the "non-efficient" authorities to verify the likely presence of serious inefficiencies in the provision of local services.

On the other hand, a green light comes on for the local authorities situated in the two upper quadrants of figure 1. Those in quadrant IV, defined as "efficient", present a potential demand that is lower than the one effectively satisfied, and standard expenditures that are greater than their actual expenditures. These local authorities should be used as benchmarks for identifying best practices. Lastly, local authorities positioned in quadrant I, defined as "over-standard" are those with actual expenditures that are higher than the standard expenditures, but that prove an actual quantity of output that is also higher than the potential demand. These local authorities should be capable of autonomously financing effective service levels that are higher than potential demand, or of reduce their service levels to bring them in line with their standard expenditures.

The joint analysis of expenditure and output in terms of gaps from the respective standard values is a new methodology, still at an experimental stage, that in the future can become a simple tool to identify local authorities target of specific policies aimed at improving efficiency in the provision of public services, as well as a simple way to identify the best practices adopted by benchmark local authorities.

SOSE aims to compute the SLS of all municipal essential services using proxies to measure the historical level of services when the outputs are not

N.B: Copia ad uso personale. È vietata la riproduzione (totale o parziale) dell'opera con qualsiasi mezzo effettuata e la sua messa a disposizione di terzi, sia in forma gratuita sia a pagamento.

easily or unambiguously measurable as, for example, in the case of local police where the quantity of services was measured considering the number of fines.

Fig. 1 – Positioning map in relation to the gap between actual expenditure and standard expenditures (horizontal axis) and between actual outputs and standard levels of services (vertical axis)



6. Conclusions

The Italian experience in measuring standard expenditure needs (SEN) of municipalities, although follows the main consolidated methodologies proposed by the international literature (such as the Regression Based Approach) and adopted also by other industrialized countries, shows some interesting innovations.

As a first innovation, the Italian government decided to integrate the information provided by official sources (Budget Sheets, National Institute of Statistics, Ministry of Education, Land Registry Office, etc.) with new data by sending all authorities a specific questionnaire for each service.

As a complementary exercise, besides the evaluation of SEN, the data collected through the questionnaires have been used to produce a system of

Copyright © FrancoAngeli

performance indicators, providing basic information on how each municipality uses its resources for the provision of the essential services. The main indicator is the gap between the standard and actual expenditure for each service, other indicators are, for example, the labour cost for employee or the share of labour costs on total expenditure or the level of services per capita. This system of indicators are integrated with a *Business Intelligence* model named *Opencivitas* (available at *www.opencivitas.it*), with the purpose of providing local authorities with an innovative online tool for informationmanagement. *Opencivitas* is primarily a benchmarking tool designed to allow each local authority to display its data and compare them with the data of other authorities with similar characteristics. The aim is to make available specific *online* management tools that allow local authorities and citizens to monitor the level of services provided and render them more efficient.

A further innovation proposed by the Italian methodology is based on the estimation of theoretical standard expenditure needs (FS) which differs from the expected values of the empirical model in that, when calculating FS, a set of target variables assume a target long term value which differs from their actual value. For instance, instead of completely recognizing the expenditure needs in relation to a local authority's labour cost, only the labour cost corresponding to the national average would be recognized. In this way, local authorities that present an effective labour cost that is lower than the national average are rewarded, and vice-versa, those whose effective labour cost is higher than the national average are penalized. When the estimation of expenditure needs is based on the cost function approach, this procedure of substituting target values for the actual values of some variables, named as *neutralization process*, is very important, since it would respond to the necessity of substituting actual output values with those deemed to be "optimal" or "essential" when calculating SEN, in order not to reproduce the path of actual expenditures in the valuation of SEN. It is important to underscore that the choice of the variables subjected to neutralization introduces an element of flexibility in the procedure for determining SEN, since the choice of the variables to be neutralized is an issue that, in some cases, can be left to policymakers' decision.

A final innovation, in conclusion, is the idea to combine the evaluation of SEN with the level of services provided by each local authority so to construct a tool which can be used to measure local governments' performance just using a simple four quadrants model which ranks local authorities according to two dimensions: first the difference between the actual expenditures and the standard expenditures (expenditure gap); second the difference between the actual level of services and the standard level of services (output gap).

N.B: Copia ad uso personale. È vietata la riproduzione (totale o parziale) dell'opera con qualsiasi mezzo effettuata e la sua messa a disposizione di terzi, sia in forma gratuita sia a pagamento.

Acknowledgement

The author would like to thank Danilo Ballanti, Bruno Chiarini, Ernesto Longobardi, Alberto Zanardi for their helpful comments. The usual caveat applies.

References

- Barabaschi N., Parlato S., Peiti C., Vignocchi C. (2014). Estimate of Standard Expenditure Needs. The Example of the Register Office. *Rivista italiana degli economisti*, 3: 353-392.
- Blöchliger H., Merk O., Charbit C., Mizell L. (2007). Fiscal Equalization in OECD Countries. OECD network on fiscal relations across levels of government, Working paper no. 4.
- Brunello G., Porcelli F., Stradiotto M. (2015). I fabbisogni standard dei comuni e delle province: un nuovo patrimonio informativo per stimolare maggiore efficienza e responsabilità nei governi locali. *Rivista di diritto finanziario e scienza delle finanze*, LXXIV, 1(I): 1-38.
- Cameron A.C., Trivedi P.K. (2010). Stata Press. In: Microeconometrics Using Stata.
- Dafflon B., Mischler P. (2007). Expenditure needs equalization at the local level: methods and practice. In Kim J., Lotz J. (Eds.). *Measuring Local Government Expenditure Needs*. The Copenhagen Workshop.
- Lockwood B., Porcelli F. (2013). Incentive Schemes for Local Government: Theory and Evidence from Comprehensive Performance Assessment. In England. *The American Economic Journal: Economic Policy*, 5(3): 1-36.
- MEF Ministry of Economics and Finance (2014a). Nota Metodologica per la determinazione dei fabbisogni standard (FC06A) Funzioni nel settore sociale al netto del servizio di asilo nido (http://www.mef.gov.it/ministero/commissioni/copaff/ fabbisogni_standard.html).
- MEF Ministry of Economics and Finance (2014b). Nota Metodologica per la determinazione dei fabbisogni standard (FC06B) Funzioni nel settore sociale servizio di asilo nido (http://www.mef.gov.it/ministero/commissioni/copaff/fabbisogni standard.html).
- OECD (1981). Measuring local expenditure needs: The Copenhagen workshop. OECD Urban Management Studies, no. 4.
- Porcelli F., Vidoli F., Dispotico R., Ballanti D. (2016). Measuring public sector performance: a four quadrants model to monitor local governments' efficiency. In: Philippopoulos A. (Ed.). *Public Sector Economics and the Need for Reforms*. MIT Press (forthcoming).
- Reschovsky A. (2007). Compensating Local Governments for Differences in Expenditure Needs in a Horizontal Fiscal Equalization Program. In: Boadway R., Shah A. (Eds.). *Intergovernmental fiscal grants: Principles and practice*, World Bank.

Copyright © FrancoAngeli

N.B: Copia ad uso personale. È vietata la riproduzione (totale o parziale) dell'opera con qualsiasi mezzo effettuata e la sua messa a disposizione di terzi, sia in forma gratuita sia a pagamento.

- SOSE (2012). Nota illustrativa di accompagnamento alle note metodologiche. Roma 28 giugno (http://www.tesoro.it/ministero/commissioni/copaff/fabbisogni_standard.html).
- SOSE (2013). Relazione riepilogativa delle determinanti dei fabbisogni standard. Roma 23 dicembre (http://www.tesoro.it/ministero/commissioni/copaff/fabbisogni standard.html).